



# Article Untethered Ultra-Wideband-Based Real-Time Locating System for Road-Worker Safety

Aitor Ochoa-de-Eribe-Landaberea <sup>1,2</sup>, Leticia Zamora-Cadenas <sup>1,2</sup> and Igone Velez <sup>1,2,\*</sup>

- <sup>1</sup> CEIT-Basque Research and Technology Alliance (BRTA), Manuel Lardizabal 15, 20018 Donostia-San Sebastian, Spain; aochoadeeribe@ceit.es (A.O.-d.-E.-L.); lzamora@ceit.es (L.Z.-C.)
- <sup>2</sup> Tecnun School of Engineering, Universidad de Navarra, Manuel Lardizabal 13,

20018 Donostia-San Sebastian, Spain

\* Correspondence: ivelez@ceit.es

Abstract: In order to reduce the accident risk in road construction and maintenance, this paper proposes a novel solution for road-worker safety based on an untethered real-time locating system (RTLS). This system tracks the location of workers in real time using ultra-wideband (UWB) technology and indicates if they are in a predefined danger zone or not, where the predefined safe zone is delimited by safety cones. Unlike previous works that focus on road-worker safety by detecting vehicles that enter into the working zone, our proposal solves the problem of distracted workers leaving the safe zone. This paper presents a simple-to-deploy safety system. Our UWB anchors do not need any cables for powering, synchronisation, or data transfer. The anchors are placed inside safety cones, which are already available in construction sites. Finally, there is no need to manually measure the positions of anchors and introduce them to the system thanks to a novel self-positioning approach. Our proposal, apart from automatically estimating the anchors' positions, also defines the limits of safe and danger zones. These features notably reduce the deployment time of the proposed safety system. Moreover, measurements show that all the proposed simplifications are obtained with an accuracy of 97%.

Keywords: accident prevention; construction sites; real-time tracking; RTLS; safety; UWB

# 1. Introduction

Occupational accidents are of major concern for workers, their families, and companies. However, according to Eurostat, over 2.8 million non-fatal accidents and 3347 fatal accidents were reported in 2021 in the European Union of 27 states (EU-27) [1]. Among all activities, the construction sector had the highest incidence rate of non-fatal accidents: 3151.9 accidents per 100,000 persons employed [2]. Thus, the construction sector has an urgent need to improve its safety proceedings.

Among all types of construction sites that exist, road construction is a particular case in which workers are exposed to serious risk of death due to heavy vehicles driving fast nearby. In order to guarantee a safe working environment, so far, the safety awareness of workers and drivers has been increased by means of personal protective equipment (PPE), informative sessions, training, and traffic control devices [3]. However, human behaviour is difficult to control [4] and the safety procedures are not always correctly followed. This risk can be avoided with the help of behaviour-based safety (BBS), which is designed to intervene and modify unsafe human behaviour. BBS is an approach of systematic application of psychological research into human behaviour. The works of [5,6] showed that, for the successful application of BBS in construction sites, constant surveillance of workers is needed. Normally, this surveillance is performed by other humans, so the collection of data is highly inefficient and can also be inaccurate due to the subjective judgments needed [7]. Thus, BBS cannot be the only approach in construction sites to improve the safety of workers.



Citation: Ochoa-de-Eribe-Landaberea, A.; Zamora-Cadenas, L.; Velez, I. Untethered Ultra-Wideband-Based Real-Time Locating System for Road-Worker Safety. *Sensors* 2024, 24, 2391. https://doi.org/10.3390/ s24082391

Academic Editor: Ren-Jye Dzeng

Received: 5 March 2024 Revised: 4 April 2024 Accepted: 6 April 2024 Published: 9 April 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

As an alternative, the application of newly emerging technologies in personalised safety monitoring could be used, a solution that has not been completely explored [8]. There are some works that have studied the feasibility of warning workers if a distracted driver enters the working zone [9,10]. Nevertheless, to our knowledge, no previous work has proposed any solution for distracted workers leaving a safe zone in a road construction environment. A good example in the case of road and railway maintenance could be the use of a real-time locating system (RTLS) that detects if workers remain in the safe working environment. However, these types of solutions may have some drawbacks because of the difficulty of deploying an RTLS [11] and the dynamic nature of construction environments [12]. Most safety systems based on RTLS [13-29] need to deploy some fixed sensors with known positions around the working environment and define safe and dangerous zones. Some time has to be spent fixing sensors where needed, measuring their positions and introducing the limits of the safe and danger zones to the system. All these tasks add preparation time to the working project. Moreover, as construction environments are highly dynamic, adjustments must be made during the lifetime of projects and the time spent preparing an RTLS is multiplied by the number of changes made. Another drawback of adding an RTLS to construction sites is that they suppose extra elements that can add difficulty to the tasks to be performed. If, for example, cables are added to power sensors, workers may trip and fall. Any safety system for construction sites should be fast and simple to use and interfere as little as possible with the work.

Apart from the general requirements of construction environments, the use case of an RTLS in road and railway maintenance also demands high accuracy. If a worker unintentionally leaves the delimited safe zone but the RTLS considers them to be in the safe zone, the worker will not receive any alarm. Thus, the worker will be at serious risk of suffering an accident. The other way around can also have negative consequences, since a worker receiving unnecessary alarms will consider the safety system unreliable and will stop paying attention to it.

This paper proposes a novel safety system for construction sites especially useful in road and railway maintenance. Our safety system is an RTLS that uses ultra-wideband (UWB) technology for the safety assessment of workers. Unlike previous works [9,10], our proposal solves the problem of accidents caused by distracted workers that leave the safe zone instead of focusing on distracted drivers that enter this safe zone. Thanks to the high accuracy of UWB technology, it can reliably tell if a worker is inside or outside a predefined safe zone. If the worker is not in the safe zone, a warning or an alarm is given, depending on the seriousness of the situation. Because of the special characteristics of construction environments, the fixed UWB sensors are placed inside safety cones and the moving sensors in the workers' hard hats. As a result, safety cones act as fixed sensors of the RTLS and safety limits at the same time, which simplifies the deployment of the system. Our proposed safety system also avoids adding congestion to construction sites, as safety cones are common in them. Moreover, because of the low power demand of UWB sensors in safety cones, they can be powered by photovoltaic cells, avoiding extra congestion by cables. Additionally, for further simplification of the deployment of the safety system, a self-positioning approach of UWB fixed sensors is proposed. This self-positioning approach, apart from estimating the positions of the anchors' coordinates, also automatically sets the limits of the safe, warning, and danger zones without needing an extra communication link among anchors.

The sections of the article are organised as follows. Section 2 reviews all the work involving RTLS for safety applications in construction environments. Section 3 describes the proposed system in depth. Section 4 describes the performed experiments for the evaluation of the safety system. Finally, Section 5 discusses the obtained results and Section 6 gives the final conclusions of the work.

## 2. State of the Art

Global navigation satellite system (GNSS) is one of the most well-known positioning systems. Nevertheless, this technology has significant limitations in indoor and urban environments [30], where many construction sites are located. Thus, since 2012, research on RTLS technologies has gained relevance for safety management in construction environments [31]. The most common technologies for this purpose are computer vision, radio frequency identification (RFID), bluetooth low energy (BLE), and UWB.

In recent years, there has been an increase in research works using computer vision in construction environments due to their abundance of visual data and the recent advancements in this area [32]. In the works of [33–35], equipment and workers were successfully tracked using computer vision. However, all these works had a common problem to deal with: occlusions. When an entity is tracked with cameras, it may temporarily hide behind an object and be lost, which leads to failure. In order to mitigate this drawback, the research works of [33–35] suggest carefully choosing the positions of cameras, so that their field of view covers as much as possible of the working place. Nevertheless, selecting adequate positions for the cameras demands some planning time and extra infrastructure to place them where desired. Moreover, the construction environment is highly dynamic, so the chosen camera positions might not be adequate during all the phases of the project.

Refs. [36,37] proposed using a combination of tracking and detection algorithms to deal with occlusions. When the track of a worker is lost, it can be detected again. However, there is still high uncertainty during the time of an occlusion in systems based on computer vision, as the position of the tracked object is completely unknown.

Radio frequency-based RTLSs are less affected by occlusions and the identification of each sensor is an easy task [16]. The latter feature is crucial if an individual alarm must be sent to the person who is detected to be in danger.

An example of radio frequency-based technologies is RFID. This technology has been intensively researched for construction environments in the last decade due to its cheap and flexible approach to identify individual people and devices [38]. RFID consists of tags and readers. Tags emit a signal which is later read by the readers to calculate the distances between sensors [39]. The main setback of RFID technology is its accuracy. The works of [40,41] proposed using RFID in construction environments, but only to send proximity alerts between workers and equipment [40] or to calculate the rough position of workers [41]. However, if the actual positions of workers need to be determined with centimetre-level accuracy, RFID is not the most suitable technology [13,42]. In the case of [42], some danger zones were defined in a construction environment with a false alarm rate of 87% if a worker stood 1 m away from one of these zones. With such a high probability of receiving false alarms, workers might end up not paying attention to the safety system and ignoring alarms.

BLE is another radio frequency-based technology that has recently gained relevance as a safety system [14–16,18,43,44]. BLE shows advantages such as its low cost and power consumption. It also has a high availability, as most smartphones use this technology. Examples of BLE in construction sites can be found in [14,15], where several sensors were placed in the construction sites of a building and a petrochemical plant, respectively. Both construction sites were full of BLE sensors; in the case of [14] 39 sensors were placed in a 39 × 27.5 m construction site, and in the case of [15], 34 sensors in a 2736 m<sup>2</sup> site. The positions of all these sensors must be measured and introduced to a software manually. Such complex systems could add significant preparation time to the construction project. Moreover, the most important drawback of BLE-based systems is their poor accuracy. Generally, BLE can have positioning errors between 1 m and 10 m [15].

The newest version of BLE, version 5.1 and 5.2, called BLE-AoA, offers enhanced accuracy. The ability to estimate the angle of arrival (AoA) improves the accuracy of these systems. In fact, the work of [16] proposed an approach where BLE-AoA is used. However, only 60% of measured estimates had submetre accuracy.

Another radio frequency-based technology for RTLSs is chirp spread spectrum (CSS). The authors of [17] proposed an RTLS for safety purposes in a construction site using CSS. The obtained positioning accuracy was 0.868 m, which is quite poor if the RTLS must determine if a worker is in a safe or danger zone.

As explained before, most systems using RFID, BLE, or CSS technologies do not always offer the necessary positioning accuracy for safety applications. As an alternative, these technologies can be combined with other sensors for enhanced accuracy. For example, the authors of [18] proposed a fusion between inertial sensors and BLE. Their system consisted of four fixed BLE sensors on tripods and tags with inertial sensors on the hard hats of workers, obtaining a mean positioning error of 0.32 m in a bridge construction site. In spite of the acceptable positioning performance, their proposal still requires a complex deployment process.

For applications of high accuracy, UWB technology is the best radio frequency-based alternative in construction environments [45]. Compared to other radio frequency-based technologies, UWB presents higher immunity to multipath effects [46], which is advantageous in environments of high congestion such as construction sites. Thanks to the high accuracy of UWB, the deployment of an RTLS can be simplified.

In the works of [19–21], different methods for the evaluation of UWB systems were defined for construction environments. The most typical unfavourable conditions of construction environments were simulated, such as the presence of construction equipment (metallic objects) in the proximity of sensors. The authors of [22] ran numerical simulations to test the applicability of UWB technology for safety systems. Some other research papers went a step further with the implementation of UWB-based RTLSs in real construction environments [23–26]. The accuracies of the RTLSs were suitable for the proposed applications, but the tested systems presented some downsides that need to be corrected. A typical UWB system, such as those of [19–26], consists of a set of UWB anchors and tags. The anchors are fixed sensors at known locations that perform time difference of arrival (TDoA) measurements with the tag. For these measurements, the anchors need an accurate clock synchronisation obtained with a wired communication. The TDoA information is then sent to a processing hub via ethernet cables to calculate the position of the tags. The first disadvantage of these systems is the use of cables, since if cables are placed incorrectly, a worker may trip and fall. There is also the possibility of accidentally damaging the cable, with the consequence of a poorer performance due to the lack of a sensor. Thus, some time has to be spent to decide the correct placement of cables. In addition to that, as the construction environments are highly dynamic, the need to change the initial placement of anchors can be very common. A tethered configuration slows these adjustments.

As a solution, different works present unterhered UWB systems for their use in construction sites [27,28] or as a safety system with a geofencing strategy [29]. However, the removal of synchronisation cables can come at the expense of poorer accuracy [28].

If good accuracy is needed without the use of a wired communication, there are other UWB-based RTLSs that use time of arrival (ToA) measurements without the need for a time synchronisation [47–49]. Nevertheless, RTLSs using ToA still present some deployment complexities that have to be overcome. The suitable positions for anchors must be selected and their positions must be measured. Moreover, if used as a safety system, the coordinates of danger zones must also be measured and introduced to the system [49].

For faster deployment, there are several works that have proposed the automatic calculation of the positions of anchors [50–53]. These works need the anchors to communicate with each other, which increases their power consumption. This increase in power demand can be acceptable if anchors are plugged in to the electrical network. In construction sites, however, it is important to avoid wired connections, so anchors should be battery-powered. Thus, reduced power consumption becomes crucial. Another setback of [50–53] is that a communication link must be added to a central computer to decide when the moment of calibration is, and when the moment of normal operation begins. The need for a central

5 of 28

computer can be burdensome, especially in a large road construction site of hundreds of metres.

A more flexible self-positioning approach is that proposed by [54], which uses a simultaneous localisation and mapping (SLAM) algorithm by fusing data of a UWB tag with an inertial measurement unit (IMU). Nevertheless, this algorithm adds computational complexity and increases the power consumption because of the IMU measurements. Moreover, if a self-positioning approach is to be used with anchors that delimit a safe zone, a polygon must be created with the positions of anchors. In this case, the order in which these positions are joined is crucial to generate the polygon. The automation of this process still remains a research gap with UWB technology.

# 3. Description of the Proposed Safety System

In this work, the proposed UWB-based safety system is intended to be used in dangerous construction environments where a safe zone must be defined for work. This is a typical case in the maintenance of roads and railways, where vehicles can be driving at high speeds close to the working place. The developed safety system can locate the workers of those environments and give them an alarm if necessary.

The architecture of the described safety system is presented in Figure 1. Let us consider a use case where maintenance work is performed on a road. In this case, workers occupy one of the lanes of the road and it becomes unavailable for normal traffic. However, vehicles are still allowed to use the other lane, so a safe working environment must be defined for workers. Usually, the safe working environment is delimited with safety cones. In this work, a safety system is proposed that adds a UWB-based RTLS to improve the safety awareness of workers. The traditional safety cones are replaced by intelligent cones, which contain UWB anchors. These anchors do not need any cables for powering them, as their low power demand allows them to be powered by photovoltaic panels. The workers wear a hard hat containing a UWB tag, as shown in Figure 2a.



**Figure 1.** Architecture of the proposed safety system. (**a**) Distance calculation between cones and tags, (**b**) operation under unsafe conditions.



Figure 2. (a) Tag on hard hat of worker and (b) deployment of safety system with marked zones.

During the operation, each pair of UWB anchor and tag communicate with each other so that the distance between them is estimated, as depicted in blue in Figure 1a. The tags collect all these data to calculate the positions of workers in real time. This information is then used to estimate if workers are in danger or not. If a worker wearing a tag is detected to be unsafe, their tag orders the closest cone to send a warning signal or an alarm, as depicted in red in Figure 1b. The type of signal depends on the severity of the situation. In the case of a warning, the closest cone to the unsafe worker turns on a light. A visual stimulus is easy to detect and non-intrusive at the same time. Thus, if a warning is given, the worker knows that they may be in danger, and at the same time, they can decide to remain where they are if they consider themselves to be safe. In the case of an alarm, the closest cone to the unsafe worker sends an acoustic signal. Depending on the situation, these signals could be difficult to see or hear. That is why the proposed system has been designed in such a way that the light is bright enough to be used on sunny days. If, in a certain situation, this is not enough, a blinking light can be used, as signals in motion are easier to detect [55]. If the work area were too noisy, a wristband could be added to the workers that sends a vibro-tactile alarm instead of an acoustic signal, as by [16,56].

The safety conditions of workers are determined by defining three different zones on the plane: the safe zone, warning zone, and danger zone. Those zones have been drawn on the photograph of Figure 2b. The area inside the polygon of cones drawn in green is the safe zone. The outer part of the polygon of cones, or the danger zone, is drawn in red. There is also a warning zone drawn in orange as wide as the base of a cone, which is close to the average shoulder width of an adult human. For the correct definition of all zones, first the positions of cones must be known by the safety system and an identifier number must be given to each cone. As illustrated in Figure 3, the positions of cones define the polygons that represent all zones. Given a cone width of  $W_c$ , two polygons are defined: polygon A and polygon B. Polygon A is the result of reducing the limits of the cones' polygon by  $W_c/2$ , while polygon B is obtained by increasing the limits of the cones' polygon by  $W_c/2$ . The area contained by polygon A is called the safe zone and the area outside polygon B is the danger zone. The area of polygon B not contained by A is the warning zone. With the definition of the warning zone, workers can be warned when they are at risk of leaving the safe zone without the necessity of giving an alarm.



Figure 3. Definition of safe, warning, and danger zones with the cones' polygon.

Even though only four cones are shown in Figure 3, the safety system supports a larger number of cones. Moreover, not all cones need to be within communication range of each other. It is only necessary that a tag be in communication range of at least four cones. It is important to note that in a road construction site, cones should be separated by about 6 m [57], a distance that is well below the communication range of a UWB system. In fact, it is even possible to combine the safety cones of our system with regular safety cones to obtain a more cost-efficient deployment.

After the deployment of the proposed safety system, it is ready to be used. The characteristics of the UWB sensors used by our safety system are shown in Table 1.

Parameter	Value	Units
Carrier frequency	6.4896	GHz
Bandwidth	499.2	MHz
Channel	5	-
Bitrate	6.8	Mbps
Pulse repetition frequency (PRF)	16	MĤz
Preamble length	128	symbols
Preamble code	3	-
Start of frame delimiter (SFD)	8	symbols
Ranging rate	5	Hz

Table 1. Configuration parameters of the ultra-wideband (UWB) sensors.

The UWB anchors and tags are from SafeLoc Systems S.L. [58] and follow the IEEE 802.15.4a standard [59]. These devices are calibrated during production by the manufacturer, so there is no need for calibration during the deployment of the safety system. The tags on the hard hats of workers communicate periodically with the anchors in the safety cones. This communication allows the estimation of the positions of workers and informs them in real time if they are working in a safe working environment or not. Depending on the situation, a warning or an alarm can be given. If a tag is in the safe zone, nothing is done, since the worker wearing that tag is working safely. If the tag is detected to be in the warning zone, a warning is given. If the tag is detected to be in the danger zone, an alarm is triggered. In the proposed system, the tag is responsible for indicating to the closest cone that a warning or danger signal should be sent. Thus, the proposed safety system does not implement multi-hop communication. However, this feature could be added if alarms or warnings are to be sent to a potential central server.

In Figure 4, a flow chart of the operation of our safety system is shown.



Figure 4. Flow chart of the operation of the safety system.

This flow chart considers the process for tag *j* at time step *i* and the subscript *n* is used to refer to any cone ranging from 0 to N - 1. First, the ranging estimates between each cone and tag  $\hat{r}_{j,n}^{(raw,i)}$  are measured by means of the two-way ranging (TWR) algorithm [59]. A time-division multiple access (TDMA) scheme is used to avoid collisions. However, ranging estimates may contain errors that have to be corrected. Thus, the same method as by [60] is used to correct the ranging estimates. In the flow chart, the corrected ranging estimates are represented as  $\hat{r}_{j,n}^{(i)}$ . The corrected ranging estimates are then used in an extended Kalman filter (EKF) algorithm as in [60] to calculate the positions of the tags  $\hat{\mathbf{p}}_{j}^{(i)}$ . In order to decide whether or not to warn a worker, it has to be determined in which zone they are. Previously, in the deployment of the safety system, two polygons have been defined based on the cones' polygon. To determine if a tag's position is inside a polygon or not, a similar algorithm to that proposed by [61] is used. If the position  $\hat{\mathbf{p}}_{j}^{(i)}$  is detected to be in the danger zone, an alarm is sent, if  $\hat{\mathbf{p}}_{j}^{(i)}$  is in the warning zone, a warning is sent, and if  $\hat{\mathbf{p}}_{i}^{(i)}$  is in the safe zone, nothing is done.

With such a configuration as in Table 1, a position estimate can have a maximum delay of 200 ms. This means that if a worker walks straight out of the safe zone, the safety system will detect that worker to be out of the safe zone 200 ms after crossing the limit. If we assume that the worker walks at 5 km/h, they will be detected 27.8 cm after leaving the safe zone. Thus, if a warning zone of about the average shoulder width of adult humans is used, the selected ranging rate of 5 Hz guarantees that the worker will receive a warning before reaching the danger zone. With the selected update rate, up to ten tags can be

monitored, which is more than enough for the number of workers usually found on many construction sites.

The resulting safety system of this work presents several advantages compared to other traditional systems. Compared to a typical safety system, where only some safety cones are deployed around the working environment, the addition of UWB sensors improves the safety awareness of workers since they are warned if they leave the safe zone. Moreover, compared to other RTLS-based safety systems, our safety system is easy to deploy, more flexible, and less intrusive. First, the use of the TWR algorithm avoids the need for any clock synchronisation among UWB anchors inside safety cones. Thus, our proposed safety system does not need any cables among cones that could interfere with the path of a worker or a piece of equipment. Second, the used UWB anchors consume around 200 mA, so photovoltaic cells can be used to power them instead of adding cables. Moreover, selecting appropriately the batteries of anchors, the system can also work during the night. The added cost of using the proposed safety system with photovoltaic cells and batteries is really affordable, especially if we compare it with the more than EUR 1 million that an accident can cost an employer [62]. Third, having the anchors of the safety system in safety cones reduces the time of deployment, since only those cones must be placed where desired, without thinking where to put the UWB anchors. The reduction in time of deployment is also crucial because of the dynamic nature of construction sites. During the lifetime of a construction project, many changes must be made, so the deployment of the safety system will be carried out several times. Thus, any reduction in preparation time is multiplied by the changes made during the project. Finally, the highly congested nature of construction environments makes it difficult to add extra material for an RTLS. However, safety cones are usually an available element on construction sites, so our proposed safety system does not occupy extra space.

#### Deployment of Safety System

The proposed safety system has an easy and simple deployment, as only safety cones must be placed where necessary and then their positions must be measured. The simplest way of measuring their positions is with a robotic total station. However, when there is a lack of such equipment, our proposed safety system adds the possibility of self-positioning cones. Thus, the deployment of the safety system remains simple even if no accurate equipment is available to measure the positions of cones.

A flow chart of the self-positioning process is shown in Figure 5.

In this process, a special tag called the calibration tag is used calculate the distances between each pair of cones. The calibration tag is represented with the subscript j = 0. First, the calibration tag is placed for a period of time T on each cone. The period of time T allows the system to obtain M ranging estimates between the calibration tag on the position of a cone and the rest of cones. The flow chart of Figure 5 considers N cones going from 0 to N - 1. The *m*th ranging estimate between cone n and the calibration tag placed on the position of cone l is represented as  $r_{0,n}^{(m,p_l)}$ . The variable m takes any value between 0 and M - 1 and the variables n and l take any value between 0 and N - 1, given that  $n \neq l$ .

Once the calibration tag has been placed on each cone, the *m*th distance estimate between cones *n* and *l*,  $d_{n,l}^{(m)}$ , is calculated as a mean of the ranging estimates from cone *n* to the calibration tag on cone *l* and from cone *l* to the calibration tag on cone *n*:

$$d_{n,l}^{(m)} = \frac{\hat{r}_{0,n}^{(m,p_l)} + \hat{r}_{0,l}^{(m,p_n)}}{2}.$$
(1)

With this process, there are *M* distance estimates between each couple of cones, which are necessary to calculate their positions. The distance estimates are saved in *M* different dissimilarity matrices:

$$\mathbf{D}^{(m)} = \begin{bmatrix} 0 & d_{0,1}^{(m)} & d_{0,2}^{(m)} & \dots & d_{0,N-1}^{(m)} \\ d_{1,0}^{(m)} & 0 & d_{1,2}^{(m)} & \dots & d_{1,N-1}^{(m)} \\ d_{2,0}^{(m)} & d_{2,1}^{(m)} & 0 & \dots & d_{2,N-1}^{(m)} \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ d_{N-1,0}^{(m)} & d_{N-1,1}^{(m)} & d_{N-1,2}^{(m)} & \dots & 0 \end{bmatrix}$$
(2)

Each dissimilarity matrix is used to calculate the positions of all cones by means of the multidimensional scaling (MDS) algorithm [63].



Figure 5. Flow chart of self-positioning method.

Finally, the positions of the cones are estimated as the mean of all results obtained with those *M* different dissimilarity matrices. However, as the MDS algorithm only determines the relative positions of the cones, the obtained results have six degrees of freedom (DoF) in space, so the different groups of cone position estimates may have different origins of coordinates. Taking into account that our proposed safety system, unlike traditional positioning systems, favours a linear growth of the infrastructure, three geometric constraints are applied:

- Constraint 1: The first cone where the calibration tag has been placed is assumed to be the origin of coordinates. Figure 6a illustrates this constraint, where cone A0 corresponds to the first cone.
- Constraint 2: The second cone where the calibration tag has been placed is assumed to only be displaced in the positive direction of the horizontal axis, i.e., x > 0, y = 0, and z = 0. Figure 6b illustrates this constraint, where cone A1 corresponds to the second cone.
- Constraint 3: The last cone where the calibration tag has been placed is assumed to have a positive y coordinate and z = 0. Figure 6c illustrates this constraint. As the example considers four cones, cone A7 corresponds to the last cone.



**Figure 6.** Application of constraints to cone positions with an example of eight cones. (**a**) Constraint 1. (**b**) Constraint 2. (**c**) Constraint 3.

Note that as shown in Figure 6, there is no need for the safety cones to form a rectangle. The proposed self-positioning method can handle any kind of polygon shape with any number of cones.

The calculated cones' positions and identifiers are stored in the calibration tag. This tag is also responsible for sending the cones' position data and identifiers to the tags on the hard hats of workers so that the operation can begin. The cones' positions and identifiers are the only information needed by the system to start working.

This proposed self-positioning process presents many advantages, especially interesting for a safety system on road construction sites. First, there is no need for communication among anchors, which reduces the power consumption. Thus, the safety system can operate with photovoltaic cells or batteries instead of plugging anchors in to a central power supply. Second, anchors do not need to be connected to a central computer that orders the start of the calibration process or the normal operation. In our case, anchors simply communicate with tags even in the calibration phase. Finally, by placing the calibration tag on each anchor, the worker easily decides in which order the positions of anchors shall be joined to generate the cones' polygon. This approach is less error-prone, as the worker does not have to manually introduce the identification number of each anchor in a central software.

## 4. Experiments

The performance of the proposed safety system was assessed with several experiments. In this section, the performed experiments as well as the used evaluation methods are described.

## 4.1. Indoor Experimental Set-Up

The proposed safety system was tested in a laboratory at the authors' institution, trying to emulate the typical case of a maintenance operation on a straight road or railway. These construction environments have a similar shape. They cover a small width of about 4 m per lane [64] or track [65] and expand along the road or railway, forming a rectangle. As shown in Figure 7a, six cones were placed, defining a rectangle of  $7 \times 4$  m.

Around the limits of that rectangle, a warning zone of 0.36 m was defined, which coincides with the width of the base of the used cones. The origin of coordinates was set on the cone In-A0 and Table 2 shows the real horizontal coordinates of all the cones.

Table 2. Real positions of cones in the indoor environment.

Cone	X (m)	Y (m)
In-A0	0.0	0.0
In-A1	4.0	0.0
In-A2	7.0	0.0
In-A3	7.0	4.0
In-A4	4.0	4.0
In-A5	0.0	4.0



Figure 7. (a) Drawing of the deployed set-up. (b) Photograph of the experimental environment.

Part of this zone was covered by an Optitrack motion capture system, which is able to track the movements of the person with millimetre-level accuracy. Because of the high accuracy of the motion capture system, it was used as ground truth to evaluate the performance of our proposed safety system.

In Figure 7b, a photograph of the experimental area is shown. The used cones with the UWB anchors inside can be seen in the photograph. There were some obstacles, such as columns or tripods to hold the Optitrack cameras, which could affect the measurements of UWB sensors. However, on a construction site, it is not expected to have a clear line of sight between sensors, so this area can give us an idea of how our proposed system could work in a real environment.

Although the real positions of the cones were measured before testing the proposed safety system, the self-positioning mode of the proposal was also used to compare how well it performs with real or estimated cone positions. The self-positioning of the safety system is performed by placing a tag on each safety cone, as explained in Section 3.

In order to test the proposed safety system, several measurements were taken. First, as shown in Figure 2a, a 1.78 m tall person wearing a hard hat with the tag walked different paths in the zones defined by the safety system. Note that informed consent was obtained from the participant for the publication of their images and the results of the experiments. The objective of these measurements was to test the positioning accuracy performance of the safety system with a moving tag. The tag on the hard hat was always oriented in the forward walking direction, while all anchors in the cones were oriented towards the inside of the polygon delimited by the anchors (cones). Thus, the relative anchor-tag angles were different during the experiments, and line-of-sight (LOS) and non-line-of-sight (NLOS) conditions occurred. The second type of measurements consisted of a person wearing a hard hat and standing at three different points close to the limits of the cones' rectangle. This was carried out to evaluate the classification performance of the safety system while guaranteeing a similar number of samples in the safe, warning, and danger zones. In Table 3, the chosen points are described. In these static experiments, the tag always faced the negative direction of the y-axis drawn in Figure 7a. Similarly to the dynamic experiments, the relative anchor-tag angles were different in each measured point. Thus, the experiment considered variability in the anchor-tag orientation.

Table 3. Positions of tested points in the indoor environment for classification performance evaluation.

Point	X (m)	Y (m)	Zone
In-P1	1.5	0.5	Safe
In-P2	1.5	0.0	Warning
In-P3	1.5	-0.5	Danger

In the experiments, the tracking of the tag was performed with the real cone positions and all the ranging and positioning estimates were saved for further analysis. The saved ranging estimates were then used in simulations along with the calculated cones' positions by the self-positioning approach. The simulations gave, as a result, the performance of the proposed safety system with the self-positioning approach. This way, the same data were used to compare the performance of the safety system with real and estimated cone positions.

# 4.2. Outdoor Experimental Set-Up

The proposed safety system was also tested in another environment to test the repeatability of the results with a different deployment. The chosen place was outside the building of the authors' institution, where a road has been adapted for pedestrian use. This way, the tests were performed on a road while guaranteeing the safety of the people taking the measurements. A picture of the set-up can be seen in Figure 8.



Figure 8. Set-up in the outdoor environment.

Similar to the indoor scenario, the real positions of cones were measured but the self-positioning approach was also used for a comparison between the proposed system with self-positioned cones and manually measured cones. The real positions of cones are shown in Table 4.

Table 4. Real positions of cones in the outdoor environment.

Cone	X (m)	Y (m)
Out-A0	0.0	0.0
Out-A1	10.0	0.0
Out-A2	20.0	0.0
Out-A3	20.0	8.0
Out-A4	10.0	8.0
Out-A5	0.0	8.0

As it can be observed, the width of the rectangle was 8 m, which could be the case in maintenance of a two-lane road. There is also a distance of 10 m between two consecutive cones in the x direction. In a real application, more regular cones could be placed in between to add a visual reference of the safety limits.

The positioning accuracy of the safety system was evaluated by placing the tag on a tripod with a height of 1.45 m at 12 different points inside and outside the zone delimited by the cones. The chosen points are presented in Table 5.

Point	X (m)	Y (m)	Zone
Out-P1	6.0	5.0	Safe
Out-P2	2.0	5.0	Safe
Out-P3	-2.0	5.0	Danger
Out-P4	-2.0	7.0	Danger
Out-P5	2.0	7.0	Safe
Out-P6	6.0	7.0	Safe
Out-P7	6.0	9.0	Danger
Out-P8	2.0	9.0	Danger
Out-P9	-2.0	9.0	Danger
Out-P10	2.0	7.5	Safe
Out-P11	2.0	8.0	Warning
Out-P12	2.0	8.5	Danger
			0

Table 5. Positions of tested points in the outdoor environment.

The last three points, Out-P10, Out-P11, and Out-P12, were also used to test the classification performance, since these are the three closest points to the limits between zones. The tag always faced the positive y direction of the local coordinate system during the measurement.

#### 4.3. Evaluation Methods

The positioning accuracy of the proposed safety system was measured by comparing the estimated positions with the real ones. In the indoor environment, the Optitrack motion capture system gave the true path, and in the outdoor environment, the predefined known points where the tag *j* was placed. For each estimated position  $\hat{\mathbf{p}}_{j}^{(i)} = \begin{bmatrix} \hat{x}_{j}^{(i)} & \hat{y}_{j}^{(i)} \end{bmatrix}^{T}$ , its error was calculated as

$$\epsilon_j^{(i)} = \sqrt{(\hat{x}_j^{(i)} - x_j^{(i)})^2 + (\hat{y}_j^{(i)} - y_j^{(i)})^2},\tag{3}$$

with  $\epsilon_j^{(i)}$  being the positioning error at time step *i*,  $\hat{x}_j^{(i)}$  and  $\hat{y}_j^{(i)}$  the two-dimensional (2D) coordinates of the estimated position, and  $x_j^{(i)}$  and  $y_j^{(i)}$  the 2D coordinates of the real position. Once all the positioning errors were calculated, the system was evaluated with the mean positioning error  $\mu$ , standard deviation of the positioning error  $\sigma$ , positioning root mean square error (RMSE) [66], and the maximum positioning error  $\epsilon_{max}$ .

The classification performance was evaluated by means of a confusion matrix [67], which is a useful tool to analyse the performance of a classifier. In the case of our proposed safety system, we evaluated how well a worker was detected to be in the safe, warning, or danger zone. As our proposed safety system considers these three different cases, a  $3 \times 3$  confusion matrix was used for its evaluation. Thus, the evaluation of the classifier was performed in three steps, considering, in each case, a positive in every step. For example, in the first step, the samples in the danger zone were considered positives, while the samples in the warning and safe zones were negatives. In the second step, the samples in the warning zone were considered positives and the samples in the danger and safe zones negatives. In the third and last step, the samples in the safe zone were considered positives and the samples in the danger and warning zones negatives. After these three steps, the total number of true positives, false positives, true negatives and false negatives were summed up to evaluate the overall performance. The accuracy and  $F_1$  score [67] were used to evaluate the performance of the classifier.

#### 5. Results

#### 5.1. Positioning Accuracy in Indoor Environment Using Real Cone Positions

The positioning accuracy of the proposed safety system was first assessed in an indoor environment. In this environment, an Optitrack system was available and was used to obtain the true path in the dynamic experiments.



In Figure 9, the walked trajectory of the indoor environment can be seen.

Figure 9. Real vs. estimated trajectory in indoor environment with real cone infrastructure.

The walked path was a rectangle around the cone In-A0 placed on the origin of coordinates. This path was walked six times, three times clockwise and three times counterclockwise. The blue points represent the measured positions by the ground truth and the red circles the estimated positions by the proposed safety system. In the same picture, the cones' positions are represented with orange squares. The corresponding metrics of the positioning error are presented in Table 6.

Table 6. Positioning accuracy of the safety system indoors with real cone infrastructure.

μ (m)	σ (m)	RMSE (m)	$\epsilon_{max}$ (m)
0.172	0.128	0.214	0.983

The columns of the table, from left to right, represent the mean positioning error  $\mu$ , the standard deviation of the positioning error  $\sigma$ , the positioning RMSE, and the maximum positioning error  $\epsilon_{max}$ . The units of all statistics are metres.

Results in Table 6 show that the position estimates are similar to those of the ground truth. The measured mean positioning error was 0.172 m with a standard deviation of 0.128 m and an RMSE of 0.214 m. The obtained maximum positioning error was 0.983 m, which happened around the bottom-left corner of the walked rectangle. At this place, the estimated position was further away from the anchors' rectangle than the real position, so an alarm was given anyway.

The positioning accuracy results suggest that the proposed safety system can accurately classify the position of a worker in the previously defined zones. The obtained RMSE was 0.214 m, notably below the average shoulder width of an adult human or the width of the base of a safety cone. Moreover, these results were obtained in spite of the obstacles in the tracking area, such as the tripods and the column, shown in Figure 7b, as well as the NLOS conditions between the tag and anchors in the cones. Note that in these measurements, there was no direct line of sight between anchor In-A4 and the tag because of the column shown in Figure 7b. Due to the realism of the tested environment, the proposed safety system is expected to classify correctly if a worker is in the danger, warning, or safe zone in a real construction environment.

## 5.2. Classification Performance in Indoor Environment Using Real Cone Positions

In order to test the classification performance of our proposed safety system, a person wearing a tag stood at three different points close to one of the limits of the cones' rectangle, as shown in Table 3. There were 537 measured samples in the danger zone, 544 samples in the warning zone, and 504 samples in the safe zone. The resulting confusion matrix is shown in Table 7.

Table 7. Confusion matrix of the safety system indoors.

		Estimated State		
		Danger	Warning	Safe
	Danger	537	0	0
Actual state	Warning	81	463	0
	Safe	0	0	504

The safety system classified the tag to be in the correct zone with an accuracy of 96.6% and an  $F_1$  score of 94.9%. The only misclassified samples were 81 samples in the warning zone detected as in danger. We consider this case safer than misclassifying the tag in the warning zone as safe, since the user of the safety system should not spend too much time in the warning zone. If a worker receives an alarm in the warning zone, they will be aware of the risk of this zone. Moreover, the warning zone had no false positives.

Note that all samples in the safe zone were classified as safe. The absence of false alarms in the safe zone is important not to annoy the worker wearing the tag and maintain their safety commitment. The other way around was also true, as all samples in the danger zone were classified as in danger. This means that an alarm was always given while the person wearing the tag remained in the danger zone, improving their safety awareness.

As a conclusion of the first two experiments in the indoor environment, the proposed safety system proved to be useful to tell with high reliability if a worker is in a safe working zone or not.

## 5.3. Indoor Performance Using Self-Positioned Cones

The proposed safety system has a simple deployment thanks to the self-positioning feature. To evaluate this feature, the calibration tag was placed on each safety cone and the distances to other cones were measured, as described in the procedure of Section 3. At the position of each cone, 20 estimates of the distance to the other cones were calculated. With a ranging rate of 5 Hz, the self-positioning demanded only 4 s per cone. The resulting cone position estimates are shown in Table 8 for the indoor environment.

As can be observed in the table, the cones were self-positioned with two different configurations. The first configuration, In-conf1, fixed the cone In-A0 as the origin of coordinates. The second configuration, In-conf2, fixed the cone In-A3 as the origin of coordinates. Two configurations are presented in this work to show the influence of the chosen origin of coordinates on the performance of the safety system. In the indoor environment, positions were always recorded close to the cone In-A0. If we fix this cone as the origin of coordinates, the positioning accuracy will not be highly influenced by the errors of the estimated cone infrastructure. However, if we fix the cone In-A3, which is the opposite of In-A0, the errors of the estimated cone infrastructure could have a higher influence.

With the estimated cone positions, a new analysis as in the previous two subsections was made to test the performance of the proposed safety system in the indoor environment. In Figure 10a, the walked real and estimated trajectories can be seen with the cone infrastructure resulting from the configuration In-conf1.

Figure 10b shows the real and estimated trajectories with the cone infrastructure resulting from the configuration In-conf2. The blue points represent the measured positions by the ground truth and the red circles the estimated positions by the proposed safety system. The real cone positions are represented with orange squares and the estimated cone positions with black squares. The corresponding statistics of the positioning error are presented in Table 9.

Conf	Como	Estimated	Estimated Position	
Conr	Cone	X (m)	Y (m)	2D Error (m)
	In-A0	0.000	0.000	0.000
	In-A1	3.963	0.000	0.037
T., (1	In-A2	6.797	-0.129	0.241
In-conf1	In-A3	6.975	3.834	0.168
	In-A4	4.101	3.894	0.146
	In-A5	0.033	3.966	0.047
	In-A0	0.106	0.022	0.108
	In-A1	4.068	0.104	0.124
I	In-A2	6.904	0.034	0.102
In-conf2	In-A3	7.000	4.000	0.000
	In-A4	4.125	4.000	0.125
	In-A5	0.057	3.988	0.058

Table 8. Real vs. estimated cone positions in indoor environment.



Figure 10. Real vs. estimated trajectory in indoor environment with estimated cone infrastructure.

Cone Positions	μ (m)	σ (m)	RMSE (m)	$\epsilon_{max}$ (m)
In-conf1	0.212	0.136	0.252	1.052
In-conf2	0.157	0.126	0.201	0.959

Table 9. Positioning accuracy of the safety system indoors with self-positioned cones.

The obtained positioning accuracy results did not highly differ from the original measurements with the real cone infrastructure. The configuration In-conf1 obtained a mean error of 0.212 m and In-conf2 obtained a mean error of 0.157 m, while the original measurement obtained a mean error of 0.172 m. The reason for these small differences between In-conf1 and In-conf2 lies in the errors that the estimated positions of cones had with different self-positioning configurations, which can be observed in the last column of Table 8. These errors influenced the estimated positions of the tag. In spite of these small differences, we can conclude that the positioning accuracy with the self-positioned cones was still good enough for the proposed application. Moreover, the RMSE value remained, in both configurations, notably below the average shoulder width of an adult human.

In order to confirm the adequacy of the proposed safety system with self-positioned cones, the classification performance was also evaluated with the same tests as in Section 5.2. Table 10 shows the obtained confusion matrices with both configurations of the self-positioning approach of cones.

Cone Positions	Actual State	Danger	Estimated State Warning	Safe
In-conf1	Danger	537	0	0
	Warning	72	472	0
	Safe	0	0	504
	Danger	537	0	0
In-conf2	Warning	71	473	0
	Safe	0	0	504

Table 10. Confusion matrix of the safety system indoors with self-positioned cones.

Both approaches obtained a good classification performance and were almost identical, with an accuracy of 97.0% and an  $F_1$  score of 95.5%. The few misclassified samples consisted of 72 samples in the warning zone misclassified as in danger in the case of In-conf1, and 71 samples in the warning zone misclassified as in danger in the case of In-conf2.

As discussed in Section 5.2, it is safer to misclassify a worker in the warning zone as in the danger zone than in the safe zone. Moreover, it is important to remark that no samples in the safe zone were misclassified as in danger and no samples in the danger zone were misclassified as safe. Moreover, both self-positioning approaches obtained a slightly better classification performance than the results with the real cone infrastructure. As a consequence, it can be concluded that using the self-positioning approach of cones at least maintains the reliability of the safety system. Not having an accurate piece of equipment to measure the cones' positions should not be a problem.

#### 5.4. Classification Performance in Movement

A key performance indicator of our proposed safety system is the number of samples in the danger zone misclassified as safe and samples in the safe zone misclassified as in danger. These misclassified samples were zero in the previous experiments, but only static measurements were taken. Taking advantage of the availability of the Optitrack system in the indoor environment, a 1.78 m tall person wearing a hard hat with a tag walked two different paths, one remaining in the safe zone and the other remaining in the danger zone. In both cases, the tag remained approximately 0.5 m away from the limits of the cones' rectangle and faced the forward walking direction. With the same two experiments, positions with the real cone infrastructure and the two configurations of the self-positioned cones were obtained. The results are shown in Figure 11.

The blue points represent the measured positions by the ground truth and the red circles the estimated positions by the proposed safety system. The real cone positions are represented with orange squares and the estimated cone positions with black squares. In Table 11, the number of correctly estimated samples is shown for each case of Figure 11.

The majority of samples were estimated to be in their correct zones. With the real cone infrastructure, shown in Figure 11a,b, only a single sample out of 246 taken in the danger zone was misclassified as in the warning zone. All 159 samples of the safe zone were classified as safe. With the estimated cone infrastructure with the configuration In-conf1, shown in Figure 11c,d, all 246 samples in the danger zone were classified as in danger. Only a single sample out of the 159 samples in the safe zone was misclassified as in the warning zone. The classification performance obtained with the estimated cone infrastructure with configuration In-conf1, shown in Figure 11e,f, was identical to that obtained with In-conf1.

As with the previous results in classification performance, no false alarms were given while the tag was in the safe zone and at least one warning was given while the tag was in the danger zone. Having a moving tag did not affect the classification performance. The classification performance was also unchanged with the real and estimated cone infrastructures.





(a) Danger zone with real cone infrastructure.





(c) Danger zone with In-conf1.



(d) Safe zone with In-conf1.



**Figure 11.** Trajectories walking 0.5 m away from the cones' limits.

Zone	Cone Infrastructure	Samples in Correct Zone	Total Samples
Danger	Real	245	246
Safe	Real	159	159
Danger	In-conf1	246	246
Safe	In-conf1	158	159
Danger	In-conf2	246	246
Safe	In-conf2	158	159

Table 11. Classification performance in movement.

### 5.5. Outdoor Performance

As the proposed safety system had a good performance in the indoor environment, its repeatability was tested by taking some more tests in an outdoor environment. As in the indoor environment, all tests were performed with the real cone infrastructure and two configurations of the self-positioning approach: Out-conf1 and Out-conf2. The estimated cone positions in the outdoor environment can be seen in Table 12.

Carl	C	Estimated	Estimated Position	
Conr	Cone	X (m)	Y (m)	2D Error (m)
	Out-A0	0.000	0.000	0.000
	Out-A1	9.966	0.000	0.034
Out confl	Out-A2	19.940	-0.306	0.312
Out-conf1	Out-A3	20.079	7.716	0.295
	Out-A4	10.007	7.889	0.111
	Out-A5	0.150	8.055	0.160
	Out-A0	0.058	-0.039	0.070
	Out-A1	10.022	0.134	0.136
Out comf?	Out-A2	20.000	0.000	0.000
Out-conf2	Out-A3	20.000	8.023	0.023
	Out-A4	9.927	8.022	0.076
	Out-A5	0.069	8.018	0.071

Table 12. Real vs. estimated cone positions in outdoor environment.

The configuration Out-conf1 fixed the cone Out-A0 to the origin of coordinates, and the configuration Out-conf2 fixed the cone Out-A2 to its real position, as it was the furthest cone from the measurement zone.

As in the indoor environment, first the positioning accuracy was tested. A tag on a tripod was placed on 12 different points inside and outside the cones' rectangle. Figure 12a,c,e show the area covered by the cones in this measurement campaign and the real and estimated positions of these cones. Figure 12b,d,f present the real and estimated positions of the tag.

The blue squares represent the real positions of the tag and the dots the estimated positions. A different colour is used for the estimated positions at each point. The orange squares represent the positions of the cones. In the cases of estimated cone infrastructure, black squares are added to indicate the estimated cone positions. The corresponding statistics of the measured positioning errors are shown in Tables 13–15.

From left to right, the columns of the table contain the following information: name of the measured point, mean positioning error  $\mu$ , standard deviation of the positioning error  $\sigma$ , positioning RMSE, and maximum positioning error  $\epsilon_{max}$ . The units of all statistics are metres.

With the ideal case of real cone infrastructure, the positioning accuracy was good enough for the application of the safety system. The mean positioning error was 0.134 m, the standard deviation was 0.069 m, the RMSE was 0.151 m, and the maximum error was



0.466 m. Moreover, all points presented an RMSE notably below the average shoulder width of an adult human.

Figure 12. Real vs. estimated measured positions of the proposed safety system outdoors.

Point	μ (m)	$\sigma$ (m)	RMSE (m)	$\epsilon_{max}$ (m)
Out-P1	0.043	0.009	0.044	0.079
Out-P2	0.113	0.013	0.114	0.225
Out-P3	0.116	0.012	0.116	0.151
Out-P4	0.196	0.023	0.197	0.277
Out-P5	0.095	0.019	0.097	0.147
Out-P6	0.203	0.015	0.204	0.254
Out-P7	0.081	0.041	0.091	0.386

0.017

0.038

0.053

0.047

0.012

0.069

0.057

0.227

0.242

0.144

0.108

0.151

0.300

0.305

0.388

0.466

0.248

0.466

Table 13. Positioning accuracy of the safety system outdoors with real cone positions.

0.054

0.224

0.237

0.136

0.107

0.134

Out-P8

Out-P9

Out-P10

Out-P11

Out-P12

All

Comparing the results in Tables 13–15, we can observe that the positioning accuracy was not significantly influenced by using the estimated cone positions instead of the real

ones. As in the indoor experiments, a satisfactory positioning accuracy was obtained both measuring the real positions of cones or using the self-positioning approach.

Point	μ (m)	$\sigma$ (m)	RMSE (m)	$\epsilon_{max}$ (m)
Out-P1	0.057	0.016	0.059	0.105
Out-P2	0.090	0.016	0.092	0.211
Out-P3	0.150	0.009	0.150	0.178
Out-P4	0.218	0.018	0.219	0.307
Out-P5	0.062	0.017	0.065	0.111
Out-P6	0.114	0.017	0.115	0.204
Out-P7	0.107	0.023	0.110	0.348
Out-P8	0.065	0.023	0.069	0.286
Out-P9	0.243	0.039	0.247	0.323
Out-P10	0.207	0.046	0.212	0.343
Out-P11	0.068	0.061	0.091	0.441
Out-P12	0.054	0.019	0.057	0.246
All	0.121	0.071	0.140	0.441

Table 14. Positioning accuracy of the safety system outdoors with configuration Out-conf1.

Table 15. Positioning accuracy of the safety system outdoors with configuration Out-conf2.

Point	μ (m)	$\sigma$ (m)	RMSE (m)	$\epsilon_{max}$ (m)
Out-P1	0.032	0.009	0.033	0.098
Out-P2	0.102	0.013	0.103	0.212
Out-P3	0.143	0.014	0.144	0.183
Out-P4	0.204	0.020	0.205	0.269
Out-P5	0.085	0.011	0.086	0.118
Out-P6	0.182	0.016	0.182	0.261
Out-P7	0.059	0.055	0.081	0.372
Out-P8	0.075	0.017	0.077	0.263
Out-P9	0.220	0.022	0.221	0.281
Out-P10	0.213	0.041	0.217	0.344
Out-P11	0.136	0.042	0.142	0.436
Out-P12	0.121	0.011	0.122	0.251
All	0.131	0.066	0.147	0.436

In order to evaluate the classification performance, the points Out-P10, Out-P11, and Out-P12 were chosen, as they were the closest ones to a limit. The resulting confusion matrix is shown in Table 16.

Cone Positions	Actual State	Danger	Estimated State Warning
	Danger	913	0
Real	Warning	62	782
	Safe	0	38
	Danger	913	0
Out-conf1	Warning	46	799
	Safe	0	0

**Table 16.** Confusion matrix of the safety system outdoors.

Danger

Warning

Safe

Out-conf2

Overall, there were 913 measured samples in the danger zone, 845 samples in the warning zone, and 943 samples in the safe zone.

913

46

0

0

799

0

0

0

943

Using the real cone positions, the classification accuracy was 97.5% with an  $F_1$  score of 96.3%. No samples in the danger zone were misclassified as safe or the other way around. Using the estimated cone positions, both approaches obtained identical results. The classification accuracy was 98.9% with an  $F_1$  score of 98.3%. Thus, we can conclude that the classification performance of our proposed safety system is similar in both outdoor and indoor environments. Moreover, our proposed safety system proved to have high reliability without the need to measure the positions of cones with extra equipment.

# 5.6. Summary of Results

In Table 17, a summary of the obtained results is shown.

_							
_	Scenario	Cone Infrastructure	μ (m)	$\sigma$ (m)	RMSE (m)	Accuracy (%)	<i>F</i> <sub>1</sub> Score (%)
_	Indoors	Real	0.172	0.128	0.214	96.6	94.9
	Indoors	In-conf1	0.212	0.136	0.252	97.0	95.5
	Indoors	In-conf2	0.157	0.126	0.201	97.0	95.5
	Outdoors	Real	0.134	0.069	0.151	97.5	96.3
	Outdoors	Out-conf1	0.121	0.071	0.140	98.9	98.3
	Outdoors	Out-conf?	0 1 3 1	0.066	0.147	98.9	98.3

Table 17. Summary of obtained results.

The proposed safety system has shown to have good accuracy with both static and dynamic measurements, so the EKF algorithm works similarly in both types of measurements. The mean positioning error and the RMSE values have always been far below the average shoulder width of an adult human. This positioning accuracy was translated to a highly reliable classification performance, which showed accuracies of at least 96.6%. If the tag was placed in the safe zone, false alarms were never registered. If the tag was in the danger zone, it was never considered to be safe. These results were obtained in two different environments with different deployments, so the results are repeatable. The self-positioning approach was also tested with the safety system, which resulted in a small impact on the classification performance. The proposed safety system using the self-positioning of cones can be as reliable as with known cone positions.

## 5.7. Comparison with State of the Art

Our proposed safety system is compared with the state of art in Table 18.

Table 18. Comparison with state-of-the-art safety systems.

Work	Test	μ (m)	RMSE (m)	Acc (%)	F <sub>1</sub> (%)	Wired Anchors	Extra Infr.	Auto- Pos.	Time to Deploy (min)
BLE [14]	Dynamic	N.A. <sup>1</sup>	N.A.	82	88	No	Yes	No	N.A.
BLE 5.1 [16]	Dynamic	N.A.	N.A.	92	84	Yes	Yes	N.A.	N.A.
BLE + IMU [18]	Dynamic	0.320	N.A.	N.A.	N.A.	No	Yes	N.A.	N.A.
CSS [17]	Static	0.868	N.A.	N.A.	N.A.	No	Yes	N.A.	13
UWB [19]	Static	0.087	N.A.	N.A.	N.A.	Yes	Yes	No	N.A.
UWB [20]	Static	N.A.	0.160	N.A.	N.A.	Yes	Yes	N.A.	N.A.
UWB [21]	Dynamic	N.A.	0.229 <sup>2</sup>	N.A.	N.A.	Yes	Yes	N.A.	N.A.
UWB [23]	Dynamic	0.908	N.A.	N.A.	N.A.	Yes	Yes	No	N.A.
UWB [24]	Static	0.15	N.A.	N.A.	N.A.	Yes	Yes	N.A.	N.A.
UWB [25]	Dynamic	0.3	N.A.	N.A.	N.A.	Yes	Yes	N.A.	N.A.
UWB [28]	Static	0.5	N.A.	N.A.	N.A.	No	Yes	No	N.A.
UWB [this work]	Static	0.129	0.146	98.4	97.6	No	No	Yes	1
UWB [this work]	Dynamic	0.180	0.223	96.9	95.3	No	No	Yes	1

<sup>1</sup> When a piece of information is missing, it is indicated with N.A. (Not available).

<sup>2</sup> Mean of RMSE values of different experiments of [21], since the overall RMSE value was not given.

The reported data in [14,16-21,23-25,28] have been used in the comparison table, as we have not been able to make this comparison by simulation or real testing. The first two columns of Table 18 contain the used technology in each work and the testing condition (static or dynamic). From the third to the sixth columns, some performance metrics are shown: the presented mean positioning error, the presented positioning RMSE, the presented classification accuracy, and the presented *F*<sub>1</sub> score. Finally, the last four columns describe the simplicity of the deployment of the proposed systems. Column 7 indicates whether or not anchors are connected by cables, Column 8 indicates whether or not extra infrastructure is needed, Column 9 indicates whether or not an auto-positioning solution of fixed sensors is proposed, and Column 10 indicates the time to deploy the safety system. Note that not all these data are available in all the works. If a piece of information is not available, this is indicated with N.A. in the corresponding cell.

Compared to the works using BLE and BLE 5.1 [14,16], our proposed safety system obtained better accuracy and  $F_1$  score. Moreover, our system is simpler to deploy. The solution of [14] needs to put the BLE fixed sensors on the ceiling of a building under construction, while [16] needs to use poles with a height of 10 m. Apart from that, after the deployment of the safety systems, the solutions of [14,16] do not mention any method to simplify the position measurement of fixed sensors.

Although the accuracy of BLE can be improved with the help of inertial sensors [18], our proposed safety system obtained better positioning accuracy. Moreover, the work of [18] is more complex to deploy than our proposal and adds extra elements to the construction site.

Compared to [17], where CSS was used, our proposed safety system obtains lower positioning mean errors thanks to UWB technology. The authors of [17] gave importance to the simplification of the deployment of their system. That is the reason why they avoided using any cabling system among fixed sensors. They also indicate that in the run experiment, two people needed about 13 minutes to set up the whole system on a construction site. Due to the differences between our tested site and theirs, it is difficult to compare the deployment time of our safety system to theirs. However, an important part of the deployment of [17] consists of moving from one point to the other of the construction site and installing the necessary fixed sensors. In the experiments presented in this work, one minute was needed to perform the deployment of the system. It should be noted that in our case, the time spent placing cones on their positions should not be counted, as safety cones would have been placed anyway. We only need to place the calibration tag, taking 4 s, on each cone to estimate the cones' positions, which considerably simplifies the deployment.

Compared to most of the works using UWB [19–21,23–25], our proposed safety system does not need any cabling system among anchors. These cabling systems suppose extra deployment time and increased hazard because of the risk for a worker to trip and fall with one of the cables. The work of [28] successfully removed the cabling system at the expense of reduced accuracy. Compared to [19–21,23–25], however, our proposed safety system achieves similar accuracy and does not need any cables among anchors.

## 6. Conclusions

In this work, a safety system based on UWB RTLS has been proposed. Our proposal is especially interesting for maintenance works of roads and railways, but it could also be used for any construction site where a safe working environment has to be delimited with cones.

Unlike other proposals, our safety system is deployed easily and adapts quickly to any change in the construction project. Moreover, our hardware is placed inside safety cones, a piece of equipment which is already common on construction sites. Thus, our proposal adds no extra congestion to construction sites and it is unobtrusive. Moreover, the high accuracy of UWB technology makes the system capable of accurately detecting if a worker is in a predetermined safe zone or not. A typical bottleneck of systems similar to ours is measuring the positions of cones that contain UWB anchors. If the needed equipment is not available to measure the positions of cones, the deployment can be burdensome and error-prone. As a solution, an approach to quickly calculate the positions of cones has been proposed using UWB measurements of available sensors. The proposed self-positioning approach does not require that anchors communicate with each other or that they be connected to a central computer. In this way, the power consumption is reduced and there is no need to add a communication link, which adds preparation time. Moreover, this novel self-positioning approach also calculates automatically the coordinates of the limits of the safe, warning, and danger zones.

The obtained results show that our proposed safety system does not have significant differences in its performance if real or estimated cone positions are used. In any case, the RMSE was always considerably below the average shoulder width of an adult human. Moreover, no false alarms were registered when the tag was in the safe zone, and in the danger zone, a warning or an alarm was always given.

Author Contributions: Conceptualization, L.Z.-C. and I.V.; methodology, A.O.-d.-E.-L., L.Z.-C. and I.V.; software, A.O.-d.-E.-L., L.Z.-C. and I.V.; validation, A.O.-d.-E.-L., L.Z.-C. and I.V.; formal analysis, A.O.-d.-E.-L., L.Z.-C. and I.V.; investigation, A.O.-d.-E.-L.; writing—original draft preparation, A.O.-d.-E.-L.; writing—review and editing, L.Z.-C. and I.V.; visualization, A.O.-d.-E.-L., L.Z.-C. and I.V.; supervision, L.Z.-C. and I.V.; project administration, L.Z.-C. and I.V.; funding acquisition, L.Z.-C. and I.V.; All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by the Basque Government under the Elkartek programme, grant number KK-2021/00082 (µ4IIOT project).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

**Data Availability Statement:** The raw data supporting the conclusions of this article will be made available by the authors on request.

**Acknowledgments:** The authors would like to thank Paul Bustamante, Javier Cejudo, and Markos Losada for their support in the preparation of the utilised hardware. Additionally, they would like to thank Juan Meléndez for his encouragement to start this research work.

Conflicts of Interest: The authors declare no conflicts of interest.

#### Abbreviations

The following abbreviations are used in this manuscript:

- 2D Two-dimensional
- AoA Angle of arrival
- BBS Behaviour-based safety
- BLE Bluetooth low energy
- CSS Chirp spread spectrum
- DoF Degrees of freedom
- EKF Extended Kalman filter
- EU-27 European Union of 27 states
- IMU Inertial measurement unit
- LOS Line of sight
- MDS Multidimensional scaling
- NLOS Non-line of sight
- PPE Personal protective equipment
- PRF Pulse repetition frequency
- RFID Radio frequency identification
- RMSE Root mean square error
- RTLS Real-time locating system
- SFD Start of frame delimiter
- SLAM Simultaneous localisation and mapping

TDMA	Time-division multiple access
TDoA	Time difference of arrival
ТоА	Time of arrival
TWR	Two way ranging
UWB	Ultra-wideband

## References

- Eurostat. Accidents at Work Statistics—Statistics Explained. 2023. Available online: https://ec.europa.eu/eurostat/statisticsexplained/index.php/Accidents\_at\_work\_statistics (accessed on 8 January 2024).
- Eurostat. Accidents at Work—Statistics by Economic Activity—Statistics Explained. 2023. Available online: https://ec.europa. eu/eurostat/statistics-explained/index.php/Accidents\_at\_work\_-\_statistics\_by\_economic\_activity (accessed on 8 January 2024).
- Kivi, A.; Olidis, C. Challenges in ensuring worker safety in active roadway work zones. In Proceedings of the 2015 Conference of the Transportation Association of Canada, Charlottetown, PE, Canada, 27–30 September 2015; pp. 27–30.
- 4. Fang, D.; Huang, Y.; Guo, H.; Lim, H.W. LCB approach for construction safety. Saf. Sci. 2020, 128, 104761. [CrossRef]
- Chen, D.; Tian, H. Behavior Based Safety for Accidents Prevention and Positive Study in China Construction Project. *Procedia* Eng. 2012, 43, 528–534. [CrossRef]
- Li, H.; Lu, M.; Hsu, S.C.; Gray, M.; Huang, T. Proactive behavior-based safety management for construction safety improvement. Saf. Sci. 2015, 75, 107–117. [CrossRef]
- Heng, L.; Shuang, D.; Skitmore, M.; Qinghua, H.; Qin, Y. Intrusion warning and assessment method for site safety enhancement. Saf. Sci. 2016, 84, 97–107. [CrossRef]
- 8. Awolusi, I.; Marks, E.; Hallowell, M. Wearable technology for personalized construction safety monitoring and trending: Review of applicable devices. *Autom. Constr.* **2018**, *85*, 96–106. [CrossRef]
- 9. Martin, J.; Rozas, A.; Araujo, A. A WSN-Based Intrusion Alarm System to Improve Safety in Road Work Zones. J. Sens. 2016, 2016, 7048141. [CrossRef]
- 10. Sakhakarmi, S.; Park, J. Improved intrusion accident management using haptic signals in roadway work zone. *J. Saf. Res.* 2022, *80*, 320–329. [CrossRef] [PubMed]
- 11. Rácz-Szabó, A.; Ruppert, T.; Bántay, L.; Löcklin, A.; Jakab, L.; Abonyi, J. Real-Time Locating System in Production Management. Sensors 2020, 20, 6766. [CrossRef] [PubMed]
- 12. Rao, A.S.; Radanovic, M.; Liu, Y.; Hu, S.; Fang, Y.; Khoshelham, K.; Palaniswami, M.; Ngo, T. Real-time monitoring of construction sites: Sensors, methods, and applications. *Autom. Constr.* **2022**, *136*, 104099. [CrossRef]
- 13. Lee, K.P.; Lee, H.S.; Park, M.; Kim, H.; Han, S. A real-time location-based construction labor safety management system. *J. Civ. Eng. Manag.* 2014, 20, 724–736. [CrossRef]
- Park, J.; Kim, K.; Cho, Y. Framework of Automated Construction-Safety Monitoring Using Cloud-Enabled BIM and BLE Mobile Tracking Sensors. J. Constr. Eng. Manag. 2016, 143, 05016019. %28ASCE%29CO.1943-7862.0001223 [CrossRef]
- 15. Zhou, C.; Li, R.; Jiang, W.; Wang, Y. Positioning precision analysis of passive phased array radar localization for construction safety monitoring in a non-line-of-sight environment via heatmap. *Measurement* **2022**, *187*, 110356. [CrossRef]
- 16. Huang, Y.; Hammad, A.; Zhu, Z. Providing proximity alerts to workers on construction sites using Bluetooth Low Energy RTLS. *Autom. Constr.* 2021, 132, 103928. [CrossRef]
- 17. Li, H.; Chan, G.; Huang, T.; Skitmore, M.; Tao, T.Y.; Luo, E.; Chung, J.; Chan, X.; Li, Y. Chirp-spread-spectrum-based real time location system for construction safety management: A case study. *Autom. Constr.* **2015**, *55*, 58–65. [CrossRef]
- Lim, J.; Song, K.; Lee, H. Real-Time Location Tracking of Multiple Construction Laborers. Sensors 2016, 16, 1869. [CrossRef]
   [PubMed]
- Saidi, K.S.; Teizer, J.; Franaszek, M.; Lytle, A.M. Static and dynamic performance evaluation of a commercially-available ultra wideband tracking system. *Autom. Constr.* 2011, 20, 519–530. [CrossRef]
- Maalek, R.; Sadeghpour, F. Accuracy assessment of Ultra-Wide Band technology in tracking static resources in indoor construction scenarios. *Autom. Constr.* 2013, 30, 170–183. [CrossRef]
- 21. Maalek, R.; Sadeghpour, F. Accuracy assessment of ultra-wide band technology in locating dynamic resources in indoor scenarios. *Autom. Constr.* **2016**, *63*, 12–26. [CrossRef]
- Andolfo, C.; Sadeghpour, F. A Probabilistic Accident Prediction Model for Construction Sites. *Procedia Eng.* 2015, 123, 15–23. [CrossRef]
- 23. Cheng, T.; Venugopal, M.; Teizer, J.; Vela, P. Performance evaluation of ultra wideband technology for construction resource location tracking in harsh environments. *Autom. Constr.* **2011**, *20*, 1173–1184. [CrossRef]
- Shahi, A.; Aryan, A.; West, J.S.; Haas, C.T.; Haas, R.C. Deterioration of UWB positioning during construction. *Autom. Constr.* 2012, 24, 72–80. [CrossRef]
- 25. Giretti, A.; Carbonari, A.; Naticchia, B.; DeGrassi, M. Design and first development of an automated real-time safety management system for construction sites. *J. Civ. Eng. Manag.* 2009, *15*, 325–336. [CrossRef]
- Carbonari, A.; Giretti, A.; Naticchia, B. A proactive system for real-time safety management in construction sites. *Autom. Constr.* 2011, 20, 686–698. Selected papers from the 26th ISARC 2009, [CrossRef]

- 27. Siddiqui, H.; Vahdatikhaki, F.; Hammad, A. Case study on application of wireless ultra-wideband technology for tracking equipment on a congested site. *J. Inf. Technol. Constr.* **2019**, *24*, 167–187.
- Cho, Y.K.; Youn, J.H.; Martinez, D. Error modeling for an untethered ultra-wideband system for construction indoor asset tracking. *Autom. Constr.* 2010, 19, 43–54. [CrossRef]
- Hodge, V.; Hawkins, R.; Hilder, J.; Habli, I. Analysing Ultra-Wide Band Positioning for Geofencing in a Safety Assurance Context. arXiv 2022, arXiv:2203.05830. https://doi.org/10.48550/ARXIV.2203.05830.
- 30. Alghisi, M.; Biagi, L. Positioning with GNSS and 5G: Analysis of Geometric Accuracy in Urban Scenarios. *Sensors* **2023**, 23, 2181. [CrossRef]
- Soltanmohammadlou, N.; Sadeghi, S.; Hon, C.K.; Mokhtarpour-Khanghah, F. Real-time locating systems and safety in construction sites: A literature review. Saf. Sci. 2019, 117, 229–242. [CrossRef]
- Asadzadeh, A.; Arashpour, M.; Li, H.; Ngo, T.; Bab-Hadiashar, A.; Rashidi, A. Sensor-based safety management. *Autom. Constr.* 2020, 113, 103128. [CrossRef]
- Yang, J.; Cheng, T.; Teizer, J.; Vela, P.; Shi, Z. A performance evaluation of vision and radio frequency tracking methods for interacting workforce. *Adv. Eng. Inform.* 2011, 25, 736–747. [CrossRef]
- Park, M.W.; Koch, C.; Brilakis, I. Three-Dimensional Tracking of Construction Resources Using an On-Site Camera System. J. Comput. Civ. Eng. 2012, 26, 541–549. %28ASCE%29CP.1943-5487.0000168 [CrossRef]
- 35. Zhu, Z.; Park, M.W.; Koch, C.; Soltani, M.; Hammad, A.; Davari, K. Predicting movements of onsite workers and mobile equipment for enhancing construction site safety. *Autom. Constr.* **2016**, *68*, 95–101. [CrossRef]
- Park, M.W.; Brilakis, I. Continuous localization of construction workers via integration of detection and tracking. *Autom. Constr.* 2016, 72, 129–142. [CrossRef]
- 37. Konstantinou, E.; Lasenby, J.; Brilakis, I. Adaptive computer vision-based 2D tracking of workers in complex environments. *Autom. Constr.* 2019, 103, 168–184. [CrossRef]
- Li, H.; Chan, G.; Wong, J.K.W.; Skitmore, M. Real-time locating systems applications in construction. *Autom. Constr.* 2016, 63, 37–47. [CrossRef]
- Ni, L.M.; Yunhao Liu.; Yiu Cho Lau.; Patil, A.P. LANDMARC: Indoor location sensing using active RFID. In Proceedings of the First IEEE International Conference on Pervasive Computing and Communications, 2003, (PerCom 2003), Fort Worth, TX, USA, 26 March 2003 ; pp. 407–415. [CrossRef]
- 40. Teizer, J. Wearable, wireless identification sensing platform: Self-Monitoring Alert and Reporting Technology for Hazard Avoidance and Training (SmartHat). J. Inf. Technol. Constr. 2015, 20, 295–312.
- 41. Costin, A.M.; Teizer, J.; Schoner, B. RFID and BIM-enabled worker location tracking to support real-time building protocol and data visualization. *J. Inf. Technol. Constr.* **2015**, *20*, 495–517.
- 42. Zhang, M.; Ghodrati, N.; Poshdar, M.; Seet, B.C.; Yongchareon, S. A construction accident prevention system based on the Internet of Things (IoT). *Saf. Sci.* 2023, *159*, 106012. [CrossRef]
- Park, J.; Cho, Y.K.; Khodabandelu, A. Sensor-Based Safety Performance Assessment of Individual Construction Workers. Sensors 2018, 18, 3897. [CrossRef]
- 44. de Gabriel, J.M.G.; Fernández-Madrigal, J.A.; López-Arquillos, A.; Rubio-Romero, J.C. Monitoring harness use in construction with BLE beacons. *Measurement* **2019**, *131*, 329–340. [CrossRef]
- Li, C.T.; Cheng, J.C.; Chen, K. Top 10 technologies for indoor positioning on construction sites. *Autom. Constr.* 2020, 118, 103309. [CrossRef]
- 46. Fontana, R.J. Recent system applications of short-pulse ultra-wideband (UWB) technology. *IEEE Trans. Microw. Theory Tech.* **2004**, 52, 2087–2104. [CrossRef]
- Kempke, B.; Pannuto, P.; Campbell, B.; Dutta, P. SurePoint: Exploiting Ultra Wideband Flooding and Diversity to Provide Robust, Scalable, High-Fidelity Indoor Localization. In Proceedings of the SenSys '16: 14th ACM Conference on Embedded Network Sensor Systems CD-ROM, Stanford, CA, USA, 14–16 November 2016; pp. 137–149. [CrossRef]
- Cao, B.; Wang, S.; Ge, S.; Liu, W. Improving the Positioning Accuracy of UWB System for Complicated Underground NLOS Environments. *IEEE Syst. J.* 2022, 16, 1808–1819. [CrossRef]
- 49. Wang, C.C.; Wang, M.; Sun, J.; Mojtahedi, M. A Safety Warning Algorithm Based on Axis Aligned Bounding Box Method to Prevent Onsite Accidents of Mobile Construction Machineries. *Sensors* **2021**, *21*, 7075. [CrossRef] [PubMed]
- Hamer, M.; D'Andrea, R. Self-Calibrating Ultra-Wideband Network Supporting Multi-Robot Localization. *IEEE Access* 2018, 6, 22292–22304. [CrossRef]
- Ridolfi, M.; Fontaine, J.; Herbruggen, B.V.; Joseph, W.; Hoebeke, J.; Poorter, E.D. UWB anchor nodes self-calibration in NLOS conditions: A machine learning and adaptive PHY error correction approach. *Wirel. Netw.* 2021, 27, 3007–3023. [CrossRef]
- Piavanini, M.; Barbieri, L.; Brambilla, M.; Cerutti, M.; Ercoli, S.; Agili, A.; Nicoli, M. A Self-Calibrating Localization Solution for Sport Applications with UWB Technology. Sensors 2022, 22, 9363. [CrossRef]
- 53. Herbruggen, B.V.; Luchie, S.; Fontaine, J.; De Poorter, E. Multihop Self-Calibration Algorithm for Ultra-Wideband (UWB) Anchor Node Positioning. *IEEE J. Indoor Seamless Position. Navig.* **2023**, *1*, 1–11. [CrossRef]
- 54. Shi, Q.; Zhao, S.; Cui, X.; Lu, M.; Jia, M. Anchor self-localization algorithm based on UWB ranging and inertial measurements. *Tsinghua Sci. Technol.* **2019**, 24, 728–737. [CrossRef]

- 55. Gros, B.L.; Greenhouse, D.S.; Cohn, T.E. Visual Warning Signals Optimized for Human Perception: What the Eye Sees Fastest. *Appl. Bionics Biomech.* **2005**, *2*, 693254. [CrossRef]
- 56. Yang, X.; Roofigari-Esfahan, N. Vibrotactile Alerting to Prevent Accidents in Highway Construction Work Zones: An Exploratory Study. *Sensors* 2023, 23, 5651. [CrossRef] [PubMed]
- 57. Traffic Safety Warehouse. What to Know about Cone Placement. 2024. Available online: https://www.trafficsafetywarehouse. com/Resources/what-to-know-about-cone-placement.asp (accessed on 28 March 2024).
- 58. SafeLoc Systems S.L. SafeLoc Systems S.L. Web Page. Available online: https://www.safeloc.es/ (accessed on 10 January 2024).
- 59. Qorvo-Decawave, 7628 Thorndike Road, Greensboro, NC, USA. DW1000 User Manual. Available online: https://www.qorvo. com/products/p/DW1000#documents (accessed on 10 January 2024).
- Zamora-Cadenas, L.; Arrue, N.; Jiménez-Irastorza, A.; Vélez, I. Improving the Performance of an FMCW Indoor Localization System by Optimizing the Ranging Estimator. In Proceedings of the 2010 6th International Conference on Wireless and Mobile Communications, Valencia, Spain, 20–25 September 2010; pp. 226–231. [CrossRef]
- 61. Hormann, K.; Agathos, A. The point in polygon problem for arbitrary polygons. Comput. Geom. 2001, 20, 131–144. [CrossRef]
- 62. Allison, R.W.; Hon, C.K.; Xia, B. Construction accidents in Australia: Evaluating the true costs. *Saf. Sci.* **2019**, *120*, 886–896. [CrossRef]
- 63. Chen, Z.X.; Wei, H.W.; Wan, Q.; Ye, S.F.; Yang, W.L. A Supplement to Multidimensional Scaling Framework for Mobile Location: A Unified View. *IEEE Trans. Signal Process.* **2009**, *57*, 2030–2034. [CrossRef]
- 64. Mecheri, S.; Rosey, F.; Lobjois, R. The effects of lane width, shoulder width, and road cross-sectional reallocation on drivers' behavioral adaptations. *Accid. Anal. Prev.* **2017**, *104*, 65–73. [CrossRef]
- 65. Trackopedia die Gleisbau-Welt-Enzyklopädie. Gleisabstände. Available online: https://www.trackopedia.com/lexikon/gleisbau-und-instandhaltung/gleisvermessung/gleisabstaende (accessed on 10 January 2024).
- 66. Rubinstein, R.Y.; Kroese, D.P. Simulation and the Monte Carlo Method, 3rd ed.; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2017.
- Han, J.; Kamber, M.; Pei, J. Classification: Basic Concepts. In *Data Mining: Concepts and Techniques*, 3rd ed.; Han, J., Kamber, M., Pei, J., Eds.; The Morgan Kaufmann Series in Data Management Systems; Morgan Kaufmann: Boston, MA, USA, 2012; pp. 327–391. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.